Author's response to reviews

Title: Short and long term treatment of asthma with intravenous nutrients

Authors:

Dr Welman A Shrader Jr., MD (wshradermd@aol.com)

Version: 2 Date: 4 May 2004

5/4/2004
Gentlemen,
Enclosed is my article "Short and long term treatment of asthma with intravenous nutrients". All of the suggested revisions have been made.
This letter is to acknowledge that I have no commercial interest in anything discussed in this article. This article does not infringe existing copyrights, or the rights of a third party. The manuscript has not been published in any journal and is not under consideration by any other journal.
Thanks for accepting this article for publication.
Sincerely,
W.A. Shrader, Jr., MD